Author's response to reviews

Title: Whole body vibration compared to conventional physiotherapy in patients with gonarthrosis: A protocol for a randomized, controlled study

Authors:

Gregor Stein (gregor.stein@uk-koeln.de)
Christina Otto (christina.otto@uk-koeln.de)
Peter Knoell (peter.knoell@uk-koeln.de)
Jan Siewe (jan.siewe@uk-koeln.de)
Thomas Kaulhausen (thomas.kaulhausen@uk-koeln.de)
Christoph Faymonville (christoph.faymonville@uk-koeln.de)
Kourosh Zarghooni (kourosh.zarghooni@uk-koeln.de)

Version: 3 Date: 10 June 2010

Author's response to reviews:

Dear ladies and gentlemen,

our manuscript should now follow the CONSORT guidelines. Especially the methods section has been adapted to it.

The files of ethical approval have been removed, ethical approval and name and number of the ethics committee have also been included in the methods section.

Sample size calculation has been included in the methods section.

The Author’s contribution section and Acknowledgments section have been placed properly and been revised.

I hope our manuscript can now proceed to the revision process. If you have any questions or remarks please do not hesitate to contact me.

With best wishes,

Gregor Stein